Author’s response to reviews

Title: Surgical management of sacral meningeal cysts by obstructing the communicating holes with muscle graft

Authors:
Kai Yang (yangkaiy@stu.xjtu.edu.cn)
Huiren Tao (huiren_tao@163.com)
Chaoshuai Feng (drfengchaoshuai@163.com)
Jiawei Xu (18729007850@163.com)
Chunguang Duan (19897550@qq.com)
Weizhou Yang (yzw_1218@163.com)
Wei Su (18092800004@163.com)
Huan Li (Lihuan816@yahoo.com)
Haopeng Li (lihaopeng3993@163.com)

Version: 2 Date: 05 Dec 2019

Author’s response to reviews:

Dear editor

Thanks for your work and consideration for our manuscript Surgical management of sacral meningeal cysts by obstructing the communicating holes with muscle graft (manuscript number: BMSD-D-19-00285). We have made corresponding revisions according to your suggestion and now submit the revised version.

As for the question you asked, the image depicted in [FIGURE #1] was originally created by our authors.

Thanks again for your work. If you need any help or information, please do not hesitate to contact us.

Haopeng Li

Department of Orthopaedics, Second Affiliated Hospital of Xi’an Jiaotong University
157th West Fifth Road, Xi’an, Shaanxi Province 710004, China
E-mail Address: lihaopeng3993@163.com

Huiren Tao
Department of Orthopaedics, Shenzhen University General Hospital
1098th Xueyuan Road of Xili University Town, Shenzhen, Guangdong Province 518055, China
E-mail Address: huiren_tao@163.com